Author's response to reviews

Title: Perceived risk, anxiety, and behavioural responses of the general public during the early phase of the Influenza A (H1N1) pandemic in the Netherlands: results of three consecutive online surveys.

Authors:

Marloes Bulits (m.bulits@ggd.rotterdam.nl)
Desirée J.M.A. Beaujean (desiree.beaujean@rivm.nl)
Onno de Zwart (o.dezwart@ggd.rotterdam.nl)
Gerjo Kok (g.kok@maastrichtuniversity.nl)
Pepijn van Empelen (p.vanempelen@erasmusmc.nl)
Jim E. van Steenbergen (jim.van.steenbergen@rivm.nl)
Jan Hendrik Richardus (j.richardus@erasmusmc.nl)
Hélène A.C.M. Voeten (h.voeten@ggd.rotterdam.nl)

Version: 4 Date: 16 December 2010

Author's response to reviews: see over
Dear Editorial Team,

As a result of your email on 13-12-2010 we have made the following changes to the format of our manuscript:

- Authors’ contributions: in the manuscript is now described that all authors read and approved the final manuscript
- Tables: in the manuscript all tables are now cited in consecutive numerical order
- Highlighting/tracking: all highlighting and tracking are removed from the manuscript
- Tables: table titles are appeared above the table and are one sentence long
- Editorial note: ‘3’ in title is changed into ‘three’
- Typography: the manuscript is thoroughly checked for typographical errors.

Next week I am out of office because of holiday (from 17 till 26 December), and I will return on Monday 27 December. If you have any questions or new versions to be checked next week, please contact my colleague Mireille Wolfers, (email m.wolfers@ggd.rotterdam.nl).

Kind regards,

Marloes Bults.

Municipal Public Health Service Rotterdam
The Netherlands
Email m.bults@ggd.rotterdam.nl